Author's response to reviews

Title: Nck2 promotes human melanoma cell proliferation, migration and invasion in vitro and primary melanoma-derived tumor growth in vivo

Authors:

Mélissa Labelle-Côté (mélissa.labelle-cote@umontreal.ca)
Julie Dusseault (jdusseault@gmail.com)
Salma Ismail (salma.ismail@mail.mcgill.ca)
Aude Picard-Cloutier (aude.picard-cloutier@mail.mcgill.ca)
Peter M. Siegel (peter.siegel@mcgill.ca)
Louise Larose (louise.larose@mcgill.ca)

Version: 3 Date: 30 September 2011

Author's response to reviews: see over
September 30, 2011

Professor Calvin Roskelley  
Associate Editor, BMC Cancer  
BioMed Central

Dear Professor Roskelley,

Please find included a revised version of our manuscript: "Nck2 promotes human melanoma cell proliferation, migration and invasion in vitro and primary melanoma-derived tumor growth in vivo" (MS: 177517132578527) according to the Editorial Production Team. Please find below a point-by-point list of changes performed in this revised version.

Sincerely yours,

Louise Larose Ph.D.  
Associate Professor  
Email: louise.larose@mcgill.ca
Point by point changes according to the Editorial Team

We provided a functional link to the website cited as reference 19.
We made sure that the references are cited in numerical order.
We removed all highlighting/tracking from the manuscript and revised it for typographical errors.
We cropped all figures as recommended by the Biomed Central online figure guideline.
We used the recommended manner to mention additional files.